Author’s response to reviews

Title: Gender and thrombolysis therapy in stroke patients with incidence dyslipidemia

Authors:

Brice Blum (bblum@email.sc.edu)
Leah Wormack (lwormack@email.sc.edu)
Mason Holtel (inathan2001@yahoo.ca)
Alexandria Penwell (apenwell@email.sc.edu)
Shyyon Lari (lari@email.sc.edu)
Brittany Walker (bawalker@email.sc.edu)
Thomas Nathaniel (nathanit@greenvillemed.sc.edu)

Version: 1 Date: 28 Aug 2018

Author’s response to reviews:

The issues listed below have been addressed in the revised submission.

Technical Comments:

1. Title page: order of authors should be the same with the system and email address of authors should be provided.

2. Consent to participate: provide the statement.

3. Consent to publish: needs clarification on the provided statement.

4. Authors Contributions: approval statement of all authors.

5. Funding: provide the role of funder.

6. Abbreviations: provide this missing section on Declaration.

Editor's comments:
After reviewing the results from the Similarity check we would like to request that you revise the discussion section of the current manuscript. Paragraphs 1, 3, 4 and 6 have over 11% similarity to your recently reported manuscript (PMID 29571851)